**Author’s response to reviews**

**Title:** Prevalence and determinants of erectile dysfunction among diabetic patients attending at hospitals of Central and Northwestern Zone of Tigray, Northern Ethiopia: a cross-sectional study

**Authors:**

Awole Seid (sawlayehu@gmail.com)

Hadgu Gerensea (hadguellen1@gmail.com)

Shambel Tarko (chombe986@gmail.com)

Yosef Zenebe (yosef.zenebe79@gmail.com)

Rahel Mezemir (Rahelmezemir7@gmail.com)

**Version:** 2  **Date:** 03 Nov 2016

**Author’s response to reviews:**

Responses to Reviewers’ Comments

Responses to reviewer # 1

1. For figure 3, kindly carry out the corrections as stated in earlier comment

Response: Admitted. Actually figure 1 & 2 are removed. There is only one figure after reviewers comment.

2. Response: Thank you for providing me additional references useful for our manuscript. They are integrated in the discussion section.

Responses to reviewer # 2

1. Kindly correct “duration of illness” to “duration of diabetes”

Response: accepted. I changed to “duration of diabetes” throughout the manuscript.